Author's response to reviews

Title: Glial heterotopia of the orbit: A rare presentation

Authors:

Ranju Sitaula (helloranju50@yahoo.com)
Gulshan Bahadur Shrestha (gulshan_shrestha@hotmail.com)
Nabin Paudel (paudel.nabin@gmail.com)
Sneha Shrestha (shresthasneha@hotmail.com)
Dev Narayan Shah (dn_shah@hotmail.com)

Version: 7 Date: 15 November 2011

Author's response to reviews: see over
**Major revisions:**

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order.

*We have done the revision as that in the submission system.*

**Minor revisions:**

Structure: Change case report to case presentation.

*We have changed.*

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

*We thoroughly went through our manuscript and have made necessary changes.*